Author's response to reviews

Title: Identification of SERPINA1 as single marker for papillary thyroid carcinoma through microarray meta analysis and quantification of its discriminatory power in independent validation

Authors:

Klemens Vierlinger (klemens.vierlinger@ait.ac.at)
Markus H Mansfeld (Markus.mansfeld@ait.ac.at)
Oskar Koperek (oskar.koperek@meduniwien.ac.at)
Christa Noehammer (christa.noehammer@ait.ac.at)
Klaus Kaserer (klaus.kaserer@meduniwien.ac.at)
Friedrich Leisch (Friedrich.Leisch@boku.ac.at)

Version: 3 Date: 17 March 2011

Author's response to reviews:

Dear Editor,

thank you for accepting our paper and please find below the required point-by-point description of the changes made to the manuscript:

- All highlighting / tracking has been removed
- Author lists on the submission system were matched with the one on the paper.
- A competing interest statement was included in the Paper
- A statement confirming that all authors read and approved the manuscript was included.
- Ordering of Figures and Tables was corrected.
- The underscore-character was removed from the link, now it works.
- Orientation of figures was changed
- One occurrence of 'Fig.' was changed to 'Figure'.